Author’s response to reviews

Title: Accelerated Tibial Fracture Union in 3rd Trimester of Pregnancy - A case report

Authors:

Mudussar A Ahmad (mudussarahmad@hotmail.com)
Damayanthi Kuhanendran (damy007@hotmail.com)
Irvine W Kamande (Irvinekamande@hotmail.com)
Charalambos Charalambides
(charalambos.charalambides@whittington.nhs.uk)

Version: 5 Date: 6 January 2008

Author’s response to reviews:

3.I think the commonest long bone to be fractured is actually the radius. This has been changed to:

"Tibial fractures are the second most common long bone fracture"